Author's response to reviews

Title: Use of selective serotonin reuptake inhibitors and risk of re-operation due to post-surgical bleeding in breast cancer patients: a Danish population-based cohort study

Authors:

Rune Gärtner (rune.gartner@rh.regionh.dk)
Deirdre Cronin-Fenton (dc@dce.au.dk)
Heidi H. Hundborg (hh@dce.au.dk)
Lars Pedersen (lap@dce.au.dk)
Timothy L. Lash (tflash@bu.edu)
Henrik Toft Sørensen (hts@dce.au.dk)
Niels Kroman (kroman@dadlnet.dk)

Version: 7 Date: 20 January 2010

Author's response to reviews:

Dear Editor,

We have made some minor edits to the paper - details below.

Empty space at page 4, line two and a typographical correction in ref. 17.

Some additional information has been added to the tables - i.e. the abbreviated drugs (SSRI, TCA) have been explained in a footnote to the tables.

Sincerely,

Henrik Toft Sørensen.